Author’s response to reviews

Title: Consumption patterns of meat, poultry, and fish after disaggregation of mixed dishes: secondary analysis of the Australian National Nutrition and Physical Activity Survey 2011-12

Authors:

Zhixian Sui (zhixian.sui@sydney.edu.au)
David Raubenheimer (david.raubenheimer@sydney.edu.au)
Anna Rangan (anna.rangan@sydney.edu.au)

Version: 2 Date: 06 Jun 2017

Author’s response to reviews:

BMC Nutrition

RE: Manuscript ID NUTN-D-17-00028

Title: Consumption patterns of meat, poultry, and fish after disaggregation of mixed dishes: secondary analysis of the Australian National Nutrition and Physical Activity Survey 2011-12

We thank you for accepting our work to be published with BMC Nutrition. We have made changes to the manuscript according to editors’ comments. We have now formatted the whole manuscript in accordance with BioMed Central editorial policies and formatting guidelines.

On behalf of the authors, thank you for considering this manuscript for publication in BMC Nutrition. We look forward to your response.

Yours sincerely

Zhixian Sui

Postdoctoral research associate, PhD, MHlSc, BHN
School of Life and Environmental Sciences
THE UNIVERSITY OF SYDNEY
Location: NE Level 4 Charles Perkins Centre | The University of Sydney | NSW | 2006
Email: zhixian.sui@sydney.edu.au